\begin{thebibliography}{10}

\bibitem{McKee}
S.~McKee, W.~A. Wulf, and T.~C. Landon.
\newblock {Bounds on Memory Bandwidth in Streamed Computations}.
\newblock In {\em {Lecture Notes in Computer Science 966}}, pages 83--99.
  Springer-Verlag, Berlin, 1995.

\bibitem{McCalpin}
John~D. McCalpin.
\newblock {Memory Bandwidth and Machine Balance in Current High Performance
  Computers}.
\newblock {\em IEEE Computer Society Technical Committee on Computer
  Architecture (TCCA) Newsletter}, December 1995.
\newblock \\http://www.computer.org/tab/tcca/news/dec95/dec95\_mccalpin.ps.

\bibitem{LINPACK1}
J.~Dongarra.
\newblock {The LINPACK Benchmark: An Explanation}.
\newblock {\em Supercomputing}, pages 10--14, Spring 1988.

\bibitem{LINPACK2}
J.~Dongarra.
\newblock {Performance of Various Computers Using Standard Linear Equations
  Software in a Fortran Environment}.
\newblock Technical Report TR MCSRD 23, Argonne National Laboratory, March
  1988.

\bibitem{Marshall97a}
J.~Marshall, C.~Hill, L.~Perelman, and A.~Adcroft.
\newblock {Hydrostatic, quasi-hydrostatic and non-hydrostatic ocean modeling}.
\newblock {\em J. Geophys. Res.}, in press 1997.

\bibitem{Klinger96}
B.~Klinger, J.~Marshall, and U.~Send.
\newblock Representation and parameterisation of deep convective plumes by
  mixing.
\newblock {\em J. Geophys. Res.}, 101 (C8):18175--18182, August 1996.

\bibitem{Speer95}
K.~G. Speer and J.~Marshall.
\newblock The growth of convective plume at seafloor hot springs.
\newblock {\em J. Marine Res.}, 53:1025--1057, 1995.

\bibitem{Visbeck96}
M.~Visbeck, J.~Marshall, and H.~Jones.
\newblock On the dynamics of convective chimneys in the ocean.
\newblock {\em J. Phys. Oceanogr.}, 26:1721--1734, September 1996.

\bibitem{Jones96}
H.~Jones and J.~Marshall.
\newblock Restratification after deep convection.
\newblock {\em J. Phys. Oceanogr.}, in press 1997.

\bibitem{Haine96}
T.~Haine and J.~Marshall.
\newblock Gravitational, symmetric and baroclinic instability of the oceanic
  mixed layer.
\newblock {\em J. Phys. Oceanogr.}, in press 1997.

\bibitem{Menem96}
D.~Menemenlis, T.~Webb, C.~Wunsch, U.~Send, and C.~Hill.
\newblock {Basin-Scale Ocean Circulation from Combined Altimetric, Tomographic
  and Model Data}.
\newblock {\em Nature}, in press 1997.

\bibitem{men97}
D.~Menemenlis and C.~Wunsch.
\newblock {Linearization of an Oceanic Circulation Model for Data Assimilation
  and Climate Studies}.
\newblock {\em Journal of Atmospheric and Oceanic Technology}, in press 1997.

\bibitem{adc96}
A.~Adcroft, C.~Hill, and J.~Marshall.
\newblock {On the Treatment of Topography in Numerical Ocean Models}.
\newblock {\em Mon. Wea. Rev.}, in press 1997.

\bibitem{Marshall97b}
J.~Marshall, A.~Adcroft, C.~Hill, and L.~Perelman.
\newblock {A finite-volume, incompressible Navier-Stokes model for studies of
  the ocean on parallel computers}.
\newblock {\em J. Geophys. Res.}, in press 1997.

\bibitem{Harlow65}
F.~H. Harlow and J.~E. Welch.
\newblock Numerical calculation of time-dependent viscous incompressible flow.
\newblock {\em Phys. Fluids}, 8:2182, 1965.

\bibitem{Brugge91}
R.~Brugge, H.~L. Jones, and J.~Marshall.
\newblock Non-hydrostatic ocean modelling for studies of open-ocean deep
  convection.
\newblock In {\em {Proceedings of the Workshop on Deep Convection and Deep
  Water Formation in the Ocean}}, pages 325--340. Elsevier Science Publishers,
  Holland, 1991.

\bibitem{Miller74b}
M.~J. Miller.
\newblock On the use of pressure as vertical co-ordinate in modelling.
\newblock {\em Quart. J. R. Meteorol. Soc.}, 100:155--162, 1974.

\bibitem{Arvind95}
A.~Shaw, K-C Cho, C.~Hill, R.~P. Johnson, J.~Marshall, and Arvind.
\newblock {A comparison of implicitly parallel multithreaded and data parallel
  implementations of an ocean model based on the Navier-Stokes equations}.
\newblock Technical Report CSG Memo 364, MIT LCS, 1995.

\bibitem{TMC91}
Thinking Machine Corporation.
\newblock {\em The CM-5 Technical Summary}, 1991.

\bibitem{Fenwick95}
D.~M.~Fenwick et.al.
\newblock {The AlphaServer 8000 Series: High-end Server Platform Development}.
\newblock {\em Digital Technical Journal}, 7(1), 1995.

\bibitem{Edmondson95}
J.~H.~Edmondson et.al.
\newblock {Internal Organization of the Alpha 21164, a 300-MHz 64-bit
  Quad-issue CMOS RISC Microprocessor}.
\newblock {\em Digital Technical Journal}, 7(1), 1995.

\bibitem{CMF}
Thinking Machine Corporation.
\newblock {\em CM Fortran Language Reference Manual, V2.1}, January 1994.

\bibitem{DEC-HPF}
J.~Harris et.al.
\newblock {Compiling High Performance Fortran for Distributed-memory Systems}.
\newblock {\em {DEC Technical Journal}}, 7(3), 1995.
\newblock \\ftp://ftp.digital.com/pub/Digital/info/DTJ/v7n3\_03\_FORTRAN.ps.

\bibitem{BLAS-DESC}
C.~Lawson, R.~Hanson, D.~Kincaid, and Krogh F.
\newblock {Basic Linear Algebra Subprograms for Fortran Usage}.
\newblock {\em ACM Transactions on Mathematical Software}, 5(3):308--323, 1979.

\bibitem{DEC-DXML}
D.~P.~Manley C.~Kamath, R.~Ho.
\newblock {DXML: A High-performance Scientific Subroutine Library}.
\newblock {\em Digital Technical Journal}, 6(3), 1994.
\newblock ftp://ftp.digital.com/pub/Digital/info/DTJ/v3n6-dxml.txt.

\bibitem{PCFD95}
C.~Hill and J.~Marshall.
\newblock {Application of a parallel Navier-Stokes model to ocean circulation}.
\newblock In {\em {Parallel Computational Fluid Dynamics: Implementations and
  Results Using Parallel Computers}}, pages 545--552. Elsevier Science
  Publishers, Holland, 1995.

\bibitem{Giering}
R.~Giering and T.~Kaminski.
\newblock {Recipes for Adjoint Code Construction}.
\newblock Technical Report 212, Max-Planck-Institute for Meteorology, 1996.

\end{thebibliography}

\section{The computer platforms}
\label{s:compare}

We compare the implementation of the MIT model on two very different
platforms: the CM-5 vector MPP, and the DEC 8400 cache-based SMP.  The CM-5
is well-suited for scientific simulation applications, and it, along with
the Cray Y-MP, is the mainstay of our day-to-day production runs.  We will
be testing a 32-processor configuration of the CM-5.

The DEC 8400 is a very high-end mainstream SMP system, designed for use in
a broad variety of both technical and non-technical situations.  Digital
has already sold several thousand 8400 systems, while in contrast, Thinking
Machines shipped less than 100 CM-5 systems.  We will be testing a 300 MHz,
4-processor configuration of the DEC 8400.

\subsection{Machine architectures}

Both systems have theoretical peak floating point performance well in
excess of 100 MFlop/s, as illustrated in Figure \ref{fig:mflop-vs-bw}.
However, the two systems have radically different architectures, which we
discuss here in more detail.

\begin{figure}
\illustrationx{3.9in}{figures/cm5.eps}
\caption{High-level architecture of the CM-5}
\label{figure:CM-5}
\end{figure}

The CM-5 MPP\cite{TMC91} was one of the first parallel computers which
could routinely out-perform traditional vector supercomputers for real
scientific problems.  The machine is built from relatively modest vector
processors connected through two communications networks, as shown in
Figure \ref{figure:CM-5}.  Each vector unit runs at 16 MHz and can issue a
multiply-add instruction each cycle, giving it a peak performance of 32
MFlop/s.  Each CM-5 node contains 4 vector units, yielding a peak
performance of 128 MFlop/s per node.
In real applications, however, sustained performance of
10-30 Mflop/s per node is more typical.

The key to exploiting the CM-5's vector units is to express computation in
terms of data-parallel vector operations, which is a natural paradigm for
regular finite difference/volume models. The vector units act as both
memory controllers and vector floating point units -- each vector unit can
read or write one 64-bit word of memory every cycle, as well as performing
one floating point operation.

Main memory on the CM-5 is distributed across the vector units and so any
sort of inter-processor communication on the CM-5 is relatively expensive,
motivating the programmer to structure codes to minimize inter-processor
communication.  Often, this entails a tradeoff of extra memory use for less
communication.

The architecture of the DEC 8400 \cite{Fenwick95} is typical of the current
generation of microprocessor-based SMP's.  As shown in Figure
\ref{figure:smp}, the DEC 8400 consists of nodes connected by a shared
system bus.  Sharing of data between the processors can only occur through
reads and writes to memory.  Each node of the 8400 consists of an Alpha
21164 microprocessor \cite{Edmondson95} with three levels of cache -- the
first two levels of cache are on-chip, and the third level of cache is
off-chip, but on the same board.  The caches are small, fast-access memories
which mirror sections of the computer's larger, slower access main memory.
Computation using data from cache is much faster than using data from main
memory.  As the caches are situated farther from the processor core, they
are successively larger and slower.

\begin{figure}
\illustrationx{3.7in}{figures/dec8400.eps}
\caption{
High-level architectures of the DEC 8400 SMP}
\label{figure:smp}
\end{figure}

Other SMP's and even successive generations of DEC SMP's may have different
sized caches, or even a different number of levels of caches.  However, all
microprocessor-based SMP's have one or more levels of cache and a shared
view of main memory.

The 8400 and other mainstream SMP's rely on caches to get good performance.  
If a memory reference from the processor is returned by the 1st-level cache, the
processor can continue working on the next cycle.  References which are
returned by the 2nd or 3rd level cache cause the processor to wait longer,
and references to the main memory can cause the processor to wait for tens
of cycles.  Each Alpha microprocessor runs at 300 MHz, and if all of the
operands are available, the Alpha can perform one floating point operation
(which may be a multiply-add) per cycle, thus having a peak performance of
600 MFlop/s.

The shared memory architecture of the 8400 allows for lower communication
costs than on the CM-5.  However, if the problem does not fit into the
8400's caches, performance may be very bad, so it is important to minimize
memory use.  It should be noted, however, that main memory itself can be
very large.

\subsection{Programming environments}

Our goal is to make a model that is accessible to the general scientist, as well
as to specialist scientific programmers and numerical modelers. A
high-level, intuitive scientific programming environment is therefore
of interest to us.

Motivated by this goal, we implemented the model on the CM-5 in
CM Fortran (CMF) \cite{CMF}, which is a high-level, data-parallel dialect
of Fortran.  For the DEC 8400, we implemented the model in 
High Performance Fortran (HPF) using Digital's native
High Performance Fortran (HPF) compiler.\cite{DEC-HPF} HPF is very similar
to CMF because the design of the HPF language standard was strongly
influenced by CMF.

HPF compilers are available for most SMP platforms, provided either by the
hardware vendor or third-party compiler vendors.  However, the quality of
HPF compilers can vary widely -- although HPF compiling technology is
fairly mature in general, some hardware vendors have not made a full
commitment in manpower and other resources to develop high-quality HPF
compilers for their platforms.  Furthermore, in our experience, third-party
compiler vendors are at a disadvantage in developing high-quality compilers
for any particular platform because (1) they must support several
platforms, and (2) details of the hardware necessary to obtain maximum
performance are usually only well-understood by the hardware vendor.

The state of affairs is improving, but any decision to move to a new
platform should be made only after evaluation of both hardware capabilities
as well as software support.
